Author's response to reviews

Title: Cyclooxygenase-2 and B-cell lymphoma-2 expression in cystitis glandularis and primary vesicle adenocarcinoma

Authors:

Zhongxing Li (wxd0422@163.com)
Guangcheng Ge (315839923@qq.com)
Rui Feng (970286944@qq.com)
Dan Wu (chao8746@126.com)
Bin Shen (drxbju@gmail.com)
Xing Wang (707035180@qq.com)
Yan Cui (amyyicss@sina.com)
Junrong Li (452478970@qq.com)
Xiaobing Ju (doctorjxb73@njmu.edu.cn)

Version: 4 Date: 22 December 2013

Author's response to reviews: see over
Dear Mr Reymon Ampalaya

Thank you for your email correspondence on 7-Nov-2013, in which you encouraged us to revise our manuscript (MS: 1288440221101870) entitled “Expression of Cyclooxygenase-2 and B-cell lymphoma-2 in cystitis glandularis and primary vesical adenocarcinoma”.

Based on your suggestions, we asked a native speaker to help with the revision of this manuscript especially for the language improvement. Now you will see a revised (highlighted) manuscript attached as a main document. We hope that the revision is acceptable, and we look forward to hearing from you soon.
Sincerely,

Xiaobing Ju, MD
Department of Urology
The First Affiliated Hospital of Nanjing Medical University
300 Guangzhou Road, Nanjing 210029, China
Tel: +86 25 83780079
Fax: +86 25 83780079
22-Dec-2013